Author's response to reviews

Title: Pattern of Traditional Medicine Use by Adult Saudi Patients with Neurological Disorders

Authors:

Yousef M Mohammad (ymohammad@ksu.edu.sa)
Ahmed N Al-Ahmari (An1410@gmail.com)
Fahad A Al-Dashash (F_a_dashash@gmail.com)
Fawaz A Al-Hussain (Faalhussain@ksu.edu.sa)
Firas F Al-Mansour (Firas_dm@hotmail.com)
Abdullah A Masoud (A_stonecold_a@hotmail.com)
Hoda a Jradi (jradiho@NGHA.MED.SA)

Version: 8
Date: 1 February 2015

Author's response to reviews: see over
Editorial Comments:

- Thank you for providing us with proof of the consent to publish patient images associated with your study. However, we do not wish to publish these alongside your manuscript, so we would now ask you to remove them from your additional files.

- Similarly, we would ask you to please include a statement in your manuscript declaring that specific consent to publish the patient images in your manuscript was obtained from the patients featured.

As requested, proof of the consent to publish patients’ images associated with our study were removed from the additional files

A statement (highlighted in yellow) was added to the first page of the manuscript, indicating a formal consent to publish the patients’ images in the manuscript was obtained from the patients featured.